Author's response to reviews

Title: Can cognitive enhancers reduce the risk of falls in older people with Mild Cognitive Impairment? A protocol for a randomised controlled double blind trial

Authors:

 Manuel Montero-Odasso (Manuel.MonteroOdasso@sjhc.london.on.ca)
 Jennie L Wells (Jennie.Wells@sjhc.london.on.ca)
 Michael J Borrie (Michael.Borrie@sjhc.london.on.ca)
 Mark Speechley (Mark.Speechley@schulich.uwo.ca)

Version: 2 Date: 10 August 2009

Author's response to reviews:

Dear sirs,

Find attached our revised version with the formatting requirements.

Sincerely,

Dr Manuel Montero-Odasso